

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bellucci, Andrea; Favaretto, Ilario; Giombini, Germana

## Working Paper Does innovation affect credit access? New empirical evidence from Italian small business lending

IAW Diskussionspapiere, No. 104

**Provided in Cooperation with:** Institute for Applied Economic Research (IAW)

*Suggested Citation:* Bellucci, Andrea; Favaretto, Ilario; Giombini, Germana (2014) : Does innovation affect credit access? New empirical evidence from Italian small business lending, IAW Diskussionspapiere, No. 104, Institut für Angewandte Wirtschaftsforschung (IAW), Tübingen

This Version is available at: https://hdl.handle.net/10419/107650

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU

# **IAW DISCUSSION PAPERS**

iaw@iaw.edu | www.iaw.edu

**IAW Diskussionspapiere** 

Nr. 104 Mai 2014 | No. 104 May 2014

Does Innovation Affect Credit Access? New Empirical Evidence from Italian Small Business Lending

Andrea Bellucci Ilario Favaretto Germana Giombini

Institut für Angewandte Wirtschaftsforschung e.V. Ob dem Himmelreich 1 | 72074 Tübingen | Germany Tel.: +49 7071 98960 | Fax: +49 7071 989699

ISSN: 1617-5654

## IAW-Diskussionspapiere

Das Institut für Angewandte Wirtschaftsforschung (IAW) Tübingen ist ein unabhängiges außeruniversitäres Forschungsinstitut, das am 17. Juli 1957 auf Initiative von Professor Dr. Hans Peter gegründet wurde. Es hat die Aufgabe, Forschungsergebnisse aus dem Gebiet der Wirtschafts- und Sozialwissenschaften auf Fragen der Wirtschaft anzuwenden. Die Tätigkeit des Instituts konzentriert sich auf empirische Wirtschaftsforschung und Politikberatung.

Dieses IAW-Diskussionspapier können Sie auch von unserer IAW-Homepage als pdf-Datei herunterladen:

http://www.iaw.edu/Publikationen/IAW-Diskussionspapiere

## ISSN 1617-5654

Weitere Publikationen des IAW:

- IAW-News (erscheinen 4x jährlich)
- IAW-Forschungsberichte

Möchten Sie regelmäßig eine unserer Publikationen erhalten? Dann wenden Sie sich bitte an uns:

IAW e.V. Tübingen, Ob dem Himmelreich 1, 72074 Tübingen, Telefon 07071 / 98 96-0 Fax 07071 / 98 96-99 E-Mail: iaw@iaw.edu

Aktuelle Informationen finden Sie auch im Internet unter:

#### http://www.iaw.edu

Der Inhalt der Beiträge in den IAW-Diskussionspapieren liegt in alleiniger Verantwortung der Autorinnen und Autoren und stellt nicht notwendigerweise die Meinung des IAW dar.

## DOES INNOVATION AFFECT CREDIT ACCESS? NEW EMPIRICAL EVIDENCE FROM ITALIAN SMALL BUSINESS LENDING.

Andrea Bellucci<sup>1\*</sup>, Ilario Favaretto<sup>2</sup>, Germana Giombini<sup>3</sup>

#### Abstract

In this paper we analyze the access to credit of innovative firms on the *price* and *non-price* dimensions of bank lending. Using information from two datasets, we use a propensity score matching procedure to estimate the impact of the innovative nature of firms on: (a) loan interest rates; (b) the probability of having to post collateral; and (c) the probability of overdrawing. Our analysis reveals that banks trade off higher interest rates and lower collateral requirements for firms involved in innovative processes. Further, innovative firms have a lower probability of being credit rationed than their non-innovative peers.

JEL Classification: D82, E43, D40, G21. Keywords: innovative firms, interest rate, firm's financing, relationship lending.

We are grateful for comments and suggestions from Giorgio Calcagnini, Alfredo Del Monte, Donato Iacobucci, Alberto Zazzaro and seminar participants at the University of Urbino, University of Naples Federico II, IAW, University of Tubingen and XXXIII Annual Conference in Regional Science (Rome). We also thank Anna Garcia Garnero and participants to the 2014 Copenhagen conference on innovation and entrepreneurship; Domenico Scalera and participants to the 2014 SIEPI annual workshop; Rebel Cole and participants to the 2014 MFA annual conference. Andrea Bellucci acknowledges the support from the FP7 Marie Curie Actions of the European Commission, via the Intra European Fellowship (Grant Agreement Number PIEF-GA-2012-331728) and previously the REPOS project. All remaining errors are our own.

<sup>&</sup>lt;sup>1</sup> IAW, Institute for Applied Economic Research, Ob dem Himmelreich 1, 72074 Tubingen, Germany. andrea.bellucci@iaw.edu \*Corresponding Author.

<sup>&</sup>lt;sup>2</sup> Università degli Studi di Urbino, Dipartimento di Economia Società Politica. Via Saffi 42 - 61029 Urbino (PU). ilario.favaretto@uniurb.it

<sup>&</sup>lt;sup>3</sup> Università degli Studi di Urbino, Dipartimento di Economia Società Politica. Via Saffi 42 - 61029 Urbino (PU). germana.giombini@uniurb.it.

#### 1. Introduction

Does innovation affect credit access? Recently, this question has been attracting attention in the economic literature, broadly limited to the issue of credit access from a firm's perspective. By contrast, the credit costs for firms with an innovative activity, i.e. the question of whether banks' lending behaviors reflect more or less prudential preferences toward the nature of these firms, and how these lending behaviors influence the pricing and supply of bank loans, remains an open question. This is especially true where small- and medium-sized innovative firms are concerned.

Due to the importance of small firms' technological activities and their dynamic role on economic growth (Solow, 1957; Act and Audretsch, 1988, 2002), in the last decade important advancements have been made in public policies to stimulate and subsidize research and development activities (R&D) (OECD, 2011). However, despite the role played by public institutions and financial systems to implement and foster investments in start-ups and development of innovative activities with the adoption of various mechanisms of evaluation and screening technologies, the R&D projects of small and medium-sized enterprises (SMEs) remain substantially difficult to finance in a freely competitive loan market. Furthermore, European SMEs, compared to large-sized firms, mainly rely on banks, while market-based financing is quantitatively small (ECB, 2009). Meanwhile, small-sized firms also usually experience access problems to bank loans as they suffer informational opaqueness problems more heavily than large-sized firms (Berger and Udell, 1998).

In the case of innovative SMEs, informational opaqueness and risk may exacerbate difficulties in the access to bank financing. In fact, for established SMEs the provision of bank financing for their R&D activities remains - in qualitative and quantitative terms - largely inefficient and constrained due to their inherent riskiness and weaknesses. For innovative firms, credit access and credit conditions may be worse for many reasons: R&D returns are intrinsically more uncertain and less predictable (Carpenter and Petersen 2002, Hall 2002); R&D appropriability (Levine et al. 2005); firms are reluctant to share information (Himmelberg and Petersen 1994); intangible capital is hardly re-deployable (Mocnik, 2001). Finally, in terms of output and consequently of expected returns, the systematic R&D undertaken by large firms is more effective in product innovation compared to the occasional R&D carried out by SMEs (Santarelli and Sterlacchini, 1990).

Furthermore, as a consequence of the recent financial crisis, the banking system has experienced a structural reform in terms of more stringent capital requirements that may further reduce the banks' propensity to engage in riskier investments, especially in a scenario involving increasingly short-term focus by investors.

Indeed, there is solid evidence emphasizing that internal and equity financing have advantages over debt (Hellmann and Stiglitz, 2000); furthermore demonstrated is how bank debt remains a less suitable source of financing for investment in innovative activities than other internal sources of financing innovative small-sized firms. Moreover an important source of banks' debt substitute is the financial support of the investments of entrepreneurial start-ups and younger innovative enterprises by venture capitalists (Robb, 2010; Brierley, 2001).

Notwithstanding this, only recently have several empirical works pointed out the importance of the role played by financial intermediation in terms the screening expertise of lenders for the provision of funds for R&D activities and, consequently, the outcomes of innovative firms' investments (Brown et al., 2012; Chava et al., 2013).

However, there is new evidence attesting to the centrality of external finance for people in the process of starting new ventures. Audretsch et all. (2009), using data on 900 nascent entrepreneurs, found that firms use their patents and prototypes in order to signal the project's feasibility and their ability to appropriate the returns from their innovation. Thus, in the case in which the financier perceives the project's signal to be credible, innovation could impact positively on external financing. They also find that the relation between finance and innovation depends on the specific stage of a start-up lifecycle. Examining the link between liquidity constraints and investment behavior for German firms of different sizes, Audretsch and Elston (2002) find that medium-sized firms appear to be more liquidity-constrained in their investment behavior than either larger or smaller sized firms suggesting that German funding infrastructures succeeded in alleviating liquidity constraints for SBEs.

In general, the aspects that may negatively affect the financing conditions of small and medium innovators, and consequently their opportunities to undertake new investments in R&D activities prevalently originate in two facts.

The first fact depends on the presence of an information gap between entrepreneurs asking for a loan and the bank managers supplying them. The informational opaqueness of small- and mediumsized innovative firms increases the uncertainty between contracting parties, reducing the lender's ability to obtain and screen the private soft information, and consequently the banks' ability of evaluating them.

Soft information, directly gathered during banks' relationship lending or indirectly through interactions with the social context of the borrower, is crucial for both contracting parties. Firstly, for the local bank managers to use the set of information on the proposed borrower's quality and/or ability to undertake the innovative project, with the aim of better discerning the nature of the riskiness involved and the possible success of the innovative investment activities proposed (Berger and Udell, 1998).

Secondly, for SMEs, which rely on the quality of this information to reduce their informational opaqueness, and increase the probability that their investments in innovative activities will be financed. Moreover, while a long lending relationship may reduce the asymmetric information problems facilitating the credit conditions for SMEs, new established firms may not have yet built such relationships (Petersen and Rajan 1995; Berger and Udell 2001), and thus suffer more from

informational opaqueness. In addition, bank financing may be limited for the R&D projects of young firms because of the higher default risk of young companies (Fritsch *et al.* 2006).<sup>4</sup>

This problem may become even more severe as the "Basle II Capital Accord" requires banks to conduct detailed risk assessment based on standardized rating systems (Czarnitzki and Hottenrott, 2011). Moreover, established firms can innovate by building on their previous inventions, e.g., product variation or improvement, while younger firms may need to conduct more fundamental R&D, which requires more resources and is much more uncertain.

The second fact that may negatively affect the lending conditions and compromise the financing of potential R&D investment refers to the risky nature of the activity of innovative firms. Like any investment, R&D "innovative" projects require a considerable amount of financial resources but, with respect to "traditional" ones, it is characterized by higher specific sunk costs for firms (which are never refundable) and by the lower availability of collateral (Hall, 2010). In addition, these fixed costs are independent of the size of the market for innovation, and consequently require a certain amount of collateralization.

Furthermore, the credit access substantially depends on a firm's characteristics and size, and many reasons explain that financing constraints may be particularly binding for smaller innovative firms. Indeed, on one side, SMEs may be disadvantaged because they cannot exploit scale economies and have fewer overall physical assets that could serve as collateral, compared to large capital-intensive companies. On the other, SMEs have to provide a higher level of collateral compared to large innovators or traditional firms, in order to mitigate moral hazard and/or adverse selection problems (Holmstrom, 1989) and attempt to get R&D investments financed. Hence, small and medium sized firms may face different conditions than more established firms.

In addition, innovative firms may be subject to negative cash flows or high return volatility, and this is especially true in the case of small innovative firms, which may have fewer possibilities to diversify their innovative projects and obtain more stable cash flows compared to large innovative firms. Indeed, small businesses generally operate with limited resources (Giudici and Paleari, 2000) and tend to redirect their efforts towards the development of a single research project that requires considerable funding relative to the turnover base. The consequence is a higher degree of project risk, in the face of uncertain returns, rendering innovative firms basically riskier than traditional firms. Another aspect refers to the fact that innovative firms typically base their activity on intellectual capital and intangible assets that amplify the problem of financial opaqueness i.e. investors find it more difficult to evaluate the profitability of their investment projects and, therefore, firms' current and future market value (Hall, 2010).

<sup>&</sup>lt;sup>4</sup> However, the regulatory reforms introduced in the Italian banking sector in the 90s fostered a pre-entry selection process with firms better equipped to reduce the "try and see" mechanism, applications. This consequently resulted in less informational opaqueness (Santarelli, 2000).

Both information opaqueness and the riskiness of innovative firm activities, by exacerbating the problem of asymmetric information between the contracting parties, increase the bank managers' efforts to assess the quality, investment riskiness and profitability of firm investments. As a consequence, such increased efforts are expected to raise the cost of obtaining financing for SMEs or limit their loan access in quantitative terms, lowering their opportunities to undertake new investments in R&D activities.

To the best of our knowledge, our work is the first to address the problem of the credit access of innovative SMEs undertaking innovative investments in price and non-price terms from a bank's side perspective, overcoming some problems of supply shock. From the point of view of firms recent literature provides evidence of financial constraints for younger and SME innovators (Petersen and Rajan, 1995; Berger and Udell, 2002; Carpenter and Petersen, 2002; Czarnitzki, 2006).

In addition, some studies using innovation survey data stressed the importance of firm innovation activity to have access to loan markets. Freel (2007) is concerned with the extent to which 'innovativeness' is associated with a lower level of loan application success by means of a sample extracted from the "Survey of Enterprise in Northern Britain" over the period 1998-2001. In general, results suggest that the most innovative firms are less successful in loan markets than their less innovative peers – though there is some variation by proxy. Moreover, there is tentative evidence that 'a little innovation may be a good thing'. Testing for financial constraints on R&D investment and how they differ from capital investment, Czarnitzki and Hottenrott (2011) show that internal constraints, measured by mark-ups, are more decisive for R&D than for capital investment. For external constraints, they find a monotonic relationship between the level of constriction and firm size for both types of investment. Using a simultaneous model and a large sample of French firms Savignac (2008) and Hajivassiliou and Savignac (2008) do not find evidence that binding financing constraints discourage innovation. With a sample of UK firms, Canepa and Stoneman (2008) find that for innovation the cost and availability of credit particularly matters for high-tech and small-sized firms.

Data on Italy suggest that the financial structure of innovative firms is oriented more toward internal sources than equity or debt (Colombo and Grilli, 2007; Magri, 2013). Calcagnini et al. (2011) analyzing financial models of innovative firms find that only a small number of the Italian firms interviewed faced problems in raising external funds for innovation, even during the period of the economic and financial crisis. Summing up the aforesaid, empirical findings concerning innovative firms' access to credit are not unique; a prevalence of empirical literature suggests that R&D investment in small or young firms may face higher capital costs and mainly be subject to binding financial constraints than larger or older firms (Czarnitzki and Hottenrott 2011).

In this study we contribute to advancing the previous empirical literature by looking at the context of small business lending to innovative firms from the point of view of the bank, and addressing two important questions. First, our analysis aims to test whether the innovative nature of firm activities is a key variable in accessing credit markets, i.e. whether banks apply different loan conditions to innovative firms in *price* and *non-price* terms. To address this research question, we estimate two models: (i) a simultaneous-equation model for the interest rate and the collateral requirements; (ii) a Probit model for the likelihood that borrowers will draw more credit than that made available by the bank. However, as argued above, tighter contract terms could be due to the higher opaqueness of innovative firms, to their higher risk, or both.

Thus, in the second part of our analysis, in order to substantiate the two arguments about the nature of differences in credit access of innovative firms – risk and informational opaqueness - we categorize the loans in our sample by dividing them into sub-samples intended to offer cross-sectional variation in the degree of borrower transparency and information asymmetry.

To analyze the credit access of small and medium-sized innovative firms we use a combination of information obtained from two datasets on firms operating in an Italian context. The first dataset contains the entire bank portfolio on more than 15,000 credit lines granted to SMEs by a major interregional Italian bank (hereafter, simply *the bank*) in two distinct years. The second dataset identifies innovative firms that carry out a narrow set of innovative activities, signaled for example, by the registration of patents with the Italian Patent Office (IPO), the European Patent Office (EPO), the United States Patent and Trademark Office (USPTO), or participation in national research projects. Eventually, we identified 95 innovative SMEs in 2004 and 109 in 2006, respectively, out of a total of more than 8000 firms.

Innovative and non-innovative firms in the dataset turn out to be different along some dimension which also explains innovation itself and, consequently, the outcome of our models, i.e. bank behaviors in terms of interest rate, collateral requirements and overdraw. Indeed, most innovative firms, for example, are larger sized, while non-innovative ones tend to be smaller, or the former businesses tend to be set up in more highly structured legal forms than the latter. This aspect is known as the 'common support problem': due to the selection processes carried out by the banks that decide how to grant credit among firms, the group of innovative firms is a special and selective one. Thus, to precisely identify the impact of innovation on loan terms, we opted to use propensity score matching (PSM). Specifically, under the matching assumptions, the only observable difference between the treated and the control group on observables is the innovative activity, and hence one can evaluate the effect of innovation on bank contract characteristics by estimating the difference in interest rates, collateral and overdraw between the treated group (innovative firms) and the matched control group (non-innovative firms).

To the best of our knowledge, this paper is the first one to empirically analyze the access to credit by innovative firms in price and non-price terms and from a bank perspective. We improve on the existing literature in two ways. First, our findings confirm theoretical arguments according to which innovative

firms have more difficulty accessing credit, finding a positive association between innovation and interest rates. This result is consistent both with an argument for information-based frictions and with the riskier nature of innovative firms compared to traditional ones. Secondly, we also find that innovative firms face less binding credit limits and thus are less likely to overdraw funds than the others. These findings together suggest that, on one hand, banks evaluate innovative firms to be riskier, and then charge them a higher price. On the other hand, however, banks also recognize that innovative firms are more profitable than other firms, and reward them with less binding credit limits (Freel, 2007).

Our analysis also shows that innovation is a key determinant for the probability that a firm will experience binding credit limits for those firms that are more likely to suffer from problems of information asymmetry. Among them, banks evaluate positively, in term of credit availability, the presence of an innovative activity carried out by the firm. Meanwhile, for those firms who are well known by the bank and for which the loan officers may rely on additional information, being innovative is not a key determinant. These findings seem to suggest that innovation may reduce the presence of credit constraints among those firms that, typically, may suffer more from financing constraints. For these firms, innovation may help to avoid the costly option of overdrawing expensive funds.

The paper is organized as follows. Section 2 describes the dataset employed, while Section 3 analyzes the model specifications. Section 4 discusses our identification strategy. Section 5 shows our empirical findings, and Section 6 concludes.

#### 2. Data and variables

To analyze innovative firms' access to credit we combine information from two datasets. The first proprietary dataset collects information on credit lines granted to small- and medium-sized enterprises (SMEs) by a large Italian bank (hereafter, simply *the bank*). The bank belongs to a major Italian banking group quoted on the Milan Stock Exchange. One of the bank's core businesses is the provision of financing to SMEs, making this dataset ideal for the purposes of our analysis. The sample includes the bank's entire portfolio of credit lines as of September 2004 and 2006, which was mainly concentrated in two Italian provinces.<sup>5</sup> The provinces are also representative of the Italian economic structure, characterized by the presence of a large number of SMEs.<sup>6</sup> Approximately 42.8% of the borrowing firms are *Sole Proprietorships*, accounting for more than 18% of the loans granted by the bank. Other legal business profiles are *Partnerships* (27.6%) and *Corporations* (26.7%) accounting for 27,7% and 43,2%

<sup>&</sup>lt;sup>5</sup> Less than 4% of the credit lines are granted to firms located in other two neighboring provinces.

<sup>&</sup>lt;sup>6</sup> Using GDP per capita in Italy in 2006 as a base of 100, both provinces are slightly richer than the average Italian province (110.3 and 101.2, respectively). In both provinces, the manufacturing sector contributes 30% of overall value added, services contribute 68%, and agriculture only 2% (for the aggregate Italian economy, these figures are 27%, 71% and 2%, respectively).

respectively of the total loans granted by the bank. The remaining 2.9% of borrowing firms take the form of *Cooperatives*. Firms operate in more than 23 industry sectors defined at the 2-digit level according to the Italian National Institute of Statistics (ISTAT).

During the timeframe of our study, the bank had more than 60 branches in both provinces. The dataset contains information on loan contract terms (e.g. credit limits, interest rates), bank-borrower relationship (e.g. length and exclusivity, whether the borrower uses other services offered by the bank, rating, collateral, portfolio, the decisional level, etc.), borrower characteristics (e.g. address, sales, industry sector, juridical form, etc.), and characteristics of the local credit market and lending branch (e.g. concentration, province, etc.). Many of these variables permit us to verify whether the relative advantage between innovative and traditional firms is determined by market concentration, by the extent of entry barriers, or by the composition of firm size within the industry (Acs and Audretsch, 1987).

The latter dataset above combines information on firm innovative activities collected by the Regional Government Agency, which aims to promote the collaboration between firms in R&D projects, facilitating their access to technological consultation and improving the knowledge of the regional innovation system. We identify the *innovative firms* within the dataset through a series of innovation activity processes in which they participated in 2004 and 2006. Examples include the registration of patents, projects funded and the involvement and development of R&D projects. A firm is defined as innovative or, more precisely, a "firm that innovates" if it pursued at least one of the activities listed in Table 1 during this period of time. Then, we construct an indicator "*Innov*", which assumes the value of 1 if the firm pursued at least one of the activities reported in Table 1, and 0 otherwise.

Following the above-mentioned criteria, overall we identify 301 innovative firms operating in these provinces. By matching information on the bank's portfolio and firm innovation activities, we identify 234 bank-innovative firm relationships, 110 in 2004 and 124 in 2006, respectively. In 2006 almost 53% of the existing innovative firms borrowed from this bank. As our focus is on small business lending, we follow the EU definition of Small Business Enterprises and exclude 15 innovative borrowers with sales exceeding €50,000,000 from our dataset<sup>7</sup>. Generally, most of the firms that carried out an innovative project in 2004 were also involved in these activities in 2006. However, in the last year available we can observe 14 firms not previously involved in innovative projects. Thus, our final sample contains 95 innovative firms in 2004 and 109 innovative ones in 2006, respectively, showing an increase of about 15%. <sup>8</sup>

<sup>&</sup>lt;sup>7</sup> The European Commission adopted the definition of Small Business Enterprises in its Recommendation of 6 May 2003 (GUCE L 124/36 del 20/05/2003).

<sup>&</sup>lt;sup>8</sup> A previous research project on innovation identified 346 innovative firms located in this region in the year 2007. Results showed that in 2007 these firms carried out at least one of the innovative activities listed in Table 1 and constituted a large share of all innovative firms located in this Region. Indeed, they represented around 58% of all residents registered in the

Table 2 reports the descriptive statistics of firms – innovative and traditional - detected in the financial dataset. Definitions and construction of the variables are described in the next section and summarized in the Appendix.

#### A. Dependent Variables

We analyze the role of innovation on price and non-price terms of the loan contract, i.e.: *Interest rates, Overdraw* and *Collateral.* 

The first dependent variable is the interest rate actually paid by the borrower for the utilized amount of credit. The average rate is 7.043%. The difference in the interest rate between innovative and traditional borrowers is, on average, statistically significant: innovative borrowers pay approximately 46 basis points more than traditional ones (6.59 vs. 7.05).

The second dependent variable we consider is *Overdraw*, a dichotomous variable taking the value of 1 if a borrower draws more credit than what is actually granted by the bank in its loan contract and 0 if he/she uses up or utilizes strictly less than the available credit line. In Italy, credit lines provide borrowers with a certain degree of flexibility by either allowing them to not fully utilize the committed credit, free of any charge, or overdraw up to a certain amount, subject to payment of fees and penalty interest rates by the overdrawing customer. Credit availability is tighter for overdrawing firms in the sense that they may count on a lower credit buffer. On average, 22% of borrowers face tight credit limits, i.e. overdraw funds. Furthermore, the frequency with which traditional firms overdraw is significantly higher with respect to innovative ones (6% vs. 22%).

The average credit limit of the contracts in our sample is  $\pounds 117,234$  with considerable variation across legal business forms ( $\pounds 49,209$  for Sole Proprietorships vs.  $\pounds 189,931$  for Corporations). The average amount of credit that is actually used is  $\pounds 72,992$ , equivalent to 62.26% of the credit limit allowed by the bank ( $\pounds 39,091$  for Sole Proprietorships vs.  $\pounds 112,468$  for Corporations – equivalent to 79.44% and 59.21%, respectively, of the credit limits set by the bank). Again, overdrawing borrowers are more likely to be located outside the local credit market (24.65% vs. 20.89%) and outside an industrial cluster (23.70% vs. 20.75%).

The third dependent variable of interest we consider is *Collateral*. It is and indicator variable, that takes the value of 1 if the credit line is secured by real collateral, and 0 otherwise. On average, 30 percent of our borrowers are required to provide collateral. This share remains stable if we consider traditional firms, while form innovative firms it decreases to 17.5 percent.

national register of research projects (universities and state-owned laboratories of research were the remaining 42%) and 50% of all patents registered with the United States Patent and Trademark Office (USPTO) between 1991 and 2007 by residents in this region (Favaretto and Zanfei, 2007). According to Favaretto and Zanfei (2007) firms were concentrated in a few and often non-front-rank innovative activities. Most of the firms (192 over 346, that is 55% of all firms) were included in the sample because they were registered with the national register of research projects.

#### B. Control Variables

The cost of credit also depends on various borrower and lender characteristics. In order to ensure that our measure of innovation does not capture factors other than innovative activity, we introduced a rich set of control variables related to borrower and lending branch characteristics and the nature of the bank-customer relationship.

*Firm Size* is measured by the borrower's total sales. As the bank only provides sales categories rather than exact sales amounts, we construct a step variable considering six categories of sales. Construction of the variable is described and summarized in the Appendix. In terms of turnover innovative firms have significantly higher figures than the traditional ones, (on average 4,88 vs. 2,22 million euro). The average borrower in our sample falls into the second category, while the innovative borrower falls into the fourth one. The expected impact of *Firm Size* is an improvement in firms' access to credit, by lowering interest rates and the likelihood of experiencing credit tightness, particularly if innovative firms are involved.

Another borrower characteristic we also consider is the legal business form of the firm. Corporation is an indicator that takes the value of 1 if the loan recipient is a corporation and 0 otherwise. In our dataset almost 34% of the borrowers were set up with a corporate form, but only 7.7% of innovative firms are not corporations. The univariate results reported in Table 2 show that innovative firms generally have a more highly structured legal business form and are mainly organized as Limited Liability companies (92% vs. 33% of traditional firms). In our sample firms that innovate have a slightly more highly structured legal business profile compared to traditional firms. This fact could be due to the industrial sector in which they operate (i.e. the secondary sector and manufacturing in particular), by the firm's size and/or by the firm's activity seniority. In the latter case it is possible that younger firms tend to be less structured (Berger and Udell, 1998). To take into consideration differences in the specialization of the borrower, we introduce a dummy variable Industrial district that takes the value of 1 if the borrower is located in an industrial cluster and 0 otherwise. From Table 2 we note that 58.5% of our borrowers are located in such clusters but on average the presence of innovative firms is higher than traditional ones (78.5% vs. 58.2% respectively). We also control for sectoral differences in the cost of credit by adding a set of industry indicators based on the ISTAT classification discussed earlier. In particular we build 23 dummies at the 2-digit ISTAT level, which roughly corresponds to the 2-digit SIC classification in the U.S.

We furthermore introduce three measures intended to capture the nature of the bank-borrower lending relationship. First, *Relationship Length* is the number of days since the firm first borrowed from our bank. On average, the firms included in the sample have been clients of the bank for 3,380 days. The average length is 3,353 days for innovative and 5,089 days for traditional firms, respectively. These figures are comparable to findings by Cole (1998), Degryse and Van Cayseele (2000) and in Italy with Gambini and Zazzaro (2012). The latter authors document the fact that firms with fewer than 50

employees maintain credit relationships with their main bank for 12 years on average. In the multivariate analysis, we take a logarithmic transformation of this variable. Second, *Multiple Lending* is a dummy variable designed to capture the exclusivity of the relationship. The variable takes the value of 1 if the firm borrows from multiple banks and 0 if the borrower has an exclusive relationship with our bank. In our sample, only 4% of the firms have an exclusive lending relationship with the bank and we do not notice differences between innovative and traditional firms. Lastly, *Other Services* considers the scope of the bank-borrower interaction (Cole and Wolken, 1995). The variable takes the value of 1 if the firm uses additional services provided by the bank and 0 otherwise. 91,1% of firms utile other financial services from this bank. However, innovative firms tend to use more of these services from the bank respect to their traditional counterparts (97,4% vs. 91%).

To take in account the organizational structure and decision making process of the bank (e.g., Liberti and Mian 2009), we include the variable *Portfolio*, that identifies the operating segment of the bank into which the borrower falls. Specifically, the variable takes the value of 1 if the bank considers the credit line to be part of its *small business market* and 0 if it is part of its *corporate market*. The small business market represents more than 90% of the bank's loan portfolio. It is important to note that the bank might consider a firm with a corporate organizational form to be part of the *small business market* because of the firm's characteristics and services it requires. This distinction has implications for the riskiness of the subset of the loan portfolio and for the internal division that manages the credit line. In our sample innovative firms fall, on average, inside corporate portfolios in 59% of the cases.

We also take into account the hierarchical level of the lending branch in the bank's organizational structure. Therefore, we construct a dummy variable, *Decisional level* that takes the value of 1 if the credit line is managed at the headquarters level and 0 if this happens at a local bank branch. In our sample only the 18% of the credit lines are managed at the headquarters level. This proportion is maintained for traditional firms, whereas the credit lines of innovative firms are managed at the headquarters level, on average, in the 58% of cases.

We also include a branch-based Herfindahl Hirschman Index (HHI), to capture local credit market conditions and competition, calculated in September 2004 and 2006. On average branches are located in contestable markets with an HHI close to 0.21. Moreover, in our formal analysis, time-invariant heterogeneity across bank branches is absorbed through branch fixed effects, while year effects account for differences in the aggregate economic conditions between 2004 and 2006.

Lastly, in order to measure the degree of collateralization of the credit line we construct a step variable *Amount Collateral* that takes the value of 1 if the credit line is not collateralized, 2 if it is collateralized up to 75% of the credit limit, and 3 if the collateral amount exceeds that 75% level. In the multivariate analysis, we use indicator variables for each category of collateralization. Specifically, D(Collateral i) takes the value of 1 if the collateralization of a credit line falls into the *i*-th category. Note

that 70% of the credit lines are not covered by collateral, i.e. fall into category 1. The fraction of credit lines in the  $2^{nd}$  and  $3^{rd}$  category is 19% and 11%, respectively.

In case of innovative firms the figures for category 1 and fixed assets and equipment are 82% and 3% respectively.

#### 3. Empirical Models

Our analysis aims to test whether the innovative nature of firms' activities is a key variable in accessing credit markets, and whether the riskiness of their activities may be reflected in the cost of credit - in *price* and *non-price* terms. As argued above, innovative firms are usually characterized by a higher level of opaqueness and more severe asymmetric information problems that could result in costly external funds. However innovative firms, as opposed to traditional ones, tend to experience rapid growth rates and the need to attract increasing amounts of external funds. Thus, in terms of access to credit, innovative firms may or may not face more difficulties than traditional ones, according to whether their weaknesses prevail over their strengths or *vice versa*.

To address our research question, we estimate two models: (i) a simultaneous-equation model for the interest rate and the collateral requirements; (ii) a Probit model for the likelihood of borrowers drawing more credit than that made available by the bank.

#### 3.1 Interest rates and Collateral: Simultaneous-equation model

The first part of our analysis aims at testing whether borrowers' innovative activities influence the loan contract in terms of interest rate and guarantee requirements. We utilize a simultaneous-equation specification that considers the relationship between the presence of collateral, interest rates and innovation. We opted for this framework because, as financial theory suggests, lenders' required return should be positively related to default risk but borrowers could compensate lenders through various combinations of interest rates and collateral (Brick and Palia, 2007 Calcagnini et al., 2014). Hence, it is possible that interest rates and collateral requirements are determined simultaneously in the contracting process. Therefore, we investigate the relationship between innovation, collateral and interest rate by estimating a set of regression equations – one for interest rate and one for collateral – in which the endogenous explanatory variable of one equation is the dependent variable for the other equation in the system. The specification of our system is outlined in equations (1A) and (1B):

$$Rate_{ijt} = c + \alpha Innov_{it} + \beta Collateral_{ijt} + \gamma Market_{jt} + \delta Bank_{it} + \eta Firm_{it} + \varepsilon_{ijt}$$
(1A)  
$$Collateral_{ijt} = c + \alpha Innov_{it} + \beta Rate_{ijt} + \gamma Market_{jt} + \delta Bank_{it} + \eta Firm_{it} + \varepsilon_{ijt}$$
(1B)

where *Rate* is the interest rate in percentage terms paid by firm *i* in market *j* at time *t* and *Collateral* is a binary dummy variable that takes the value of 1 if firm *i* posts collateral and 0 otherwise. *Innov* is the dummy variable capturing the innovative firms. *Market, Bank* and *Firm* are vectors of characteristics related to the local credit market (concentration, industrial district, population, etc.), the bank-firm relationship (credit limit, services, exclusivity of relationship, length, riskiness, rating, portfolio, decisional level, etc.), and firm profile (sales, activity sector, organizational form, etc.). We also include industry, branch and year-specific effects.

If banks evaluate the innovative nature of firm activities to be riskier, interest rates should be higher for innovative firms. In contrast, if banks consider innovative firms more profitable than traditional ones, then the interest rates charged to innovative firms could be lower than those charged to the other firms. Finally, interest rates should be the same for both firms (e.g. innovative and traditional ones) if banks do not evaluate innovative projects or activities. A positive point estimate of a in equation (1A) above would be consistent with the hypothesis that lending to innovative firm is riskier. Tighter contract terms on loans to innovative firms could also be reflected in a higher likelihood of having to pledge collateral compared to traditional firms. In this case, the estimated coefficient of *Innov* in model (1B) below should be positive (a > 0). Alternatively, banks may ask for lower collateral requirements to offset higher interest rates, and in this case the estimated coefficient of *Innov* in model (1B) below should be negative (a < 0).

Equations (1A) and (1B) explicitly show that *Collateral* and *I\_Rate* are determined simultaneously, i.e.: we explain interest rates with collateral but collateral is also explained by interest rates and other variables, and is not exogenous anymore. This result in endogeneity, i.e. *Collateral (I\_Rate)* is correlated with the error-term, and therefore OLS will lead to biased and inconsistent estimates. Thus, we estimate the system of equations (1A)-(1B) by means of a Three Stage Least Square regression (3-SLS). Differently from 2-SLS, 3-SLS uses the additional information that these equations could be related through their error-terms, thus, increasing the efficiency of the results (Zellner and Theil ,1962). Specifically, we adopt *I\_Rate* and *Collateral* as endogenous variables and we use the *Rival Distance*, *Rating*, and *Overdraw* as additional exogenous variables; the latter are not included in any of the system equations (1A)-(1B) - These variables will be added to the exogenous variables in the system and used in the first stage as instruments for the endogenous variables.

#### 3.2 Innovation and credit availability

Lastly, as banks may trade off easier access in exchange for higher credit prices, we focus on the quantity dimension of the loan contract. However, our dataset does not provide information on access, and all firms received LCs. Thus, as explained, we focus on over-drafting after receiving credit.<sup>9</sup>

The underlying rationale is that tighter contract terms on loans for the activities of innovative enterprises would indicate that innovative borrowers are on average riskier, and consequently there could also be implications in terms of credit rationing and availability, and for the likelihood of overdrawing. To investigate, we estimate a Probit model for the probability that a borrower used credit in excess of what is made available by the bank, i.e. overdrew funds, as follows:

$$Pt(Overdraw=1)_{ijt} = F\left(c + \alpha Innov_{it} + \gamma Market_{jt} + \delta Bank_{it} + \eta Firm_{it} + \varepsilon_{ijt}\right)$$
(2)

where *Overdraw* is an indicator that takes the value of 1 if a borrower exceeds the credit limit on the loan contract, and 0 otherwise. F(.) is the cdf of the standard normal distribution. The innovation variable *Innov*, and the vectors of controls for market conditions, bank-firm relationship, and firm characteristics (*Market, Bank*, and *Firm*, respectively) are the same as those outlined in model (1).

If innovation proxies for riskiness, borrowers dealing with innovation activities should face more binding credit limits and thus should be more likely to overdraw funds. Hence, a positive coefficient ( $\alpha > 0$ ) on the measure of innovation would be consistent with lending practices of constraint. By contrast, a negative coefficient ( $\alpha < 0$ ) on the innovation variable should reflect the fact that banks evaluate the innovation activity pursued positively.

#### 4. Identification: propensity score matching results

The aim of this study is to verify whether innovation is a key variable of interest for lenders, and whether banks apply different conditions on loans granted to innovative and non-innovative firms, i.e. we would like to identify the effect of innovation on loan contract characteristics. Ideally, one should need innovative and non-innovative firms to be identical along every possible dimension other than innovation. Unfortunately, this assumption does not hold in our dataset, as very different types of firm are included in both groups, and innovative and non-innovative firms turn out to be significantly

<sup>&</sup>lt;sup>9</sup> A concern in using overdraw regards the fact that it is different from not receiving credit at all, but our dataset contains information on interest rates and contract terms only for successful firms: denied firms are excluded, as are firms that chose not to apply. As a consequence, there might be a survivorship bias involved. Indeed, no credit leads to firm failure (Cole, 1998). Analysis of U.S. firms using 2003 SSBF by Cole (2010) shows that denial rate for renewals is an order of magnitude less than for new LCs, suggesting that is it likely that loan terms also differ across new LCs vs. renewals of LCs.

different, as shown from the descriptive statistics in Table 2. Indeed, there were 95 innovative SMEs in 2004 and 109 in 2006, respectively, out of a total of more than 8000 firms.

Thus, we need a method to solve the problem that innovative and non-innovative firms are different along some dimension, which also explains innovation itself and, consequently, the outcome of our models, i.e. bank behaviors in terms of interest rate, collateral requirements and overdraw. Indeed, for example, most innovative firms are larger while non-innovative ones tend to be smaller, or the former tend to be organized in a more highly structured business legal form than the latter. This aspect is known as the 'common support problem': due to selection processes on the part of the banks that decide how to grant credit among firms, the group of innovative firms is a special and selective one.

Thus, we opted to use propensity score matching (PSM), as has been done by, e.g. Almus and Czarnitzki (2003) or Gorg and Strobl (2007), whose research specifically deals with the potential common support problem in analyzing the impact of R&D subsidies on innovation. Under the matching assumptions, the only difference between the treated and the control group on observables is innovative activity, and hence one can evaluate the effect of innovation on bank contract characteristics by estimating the difference in interest rates, collateral and overdraw between the treated group (innovative firms) and the matched control group (non-innovative firms).

One crucial assumption of this approach, though, is unconfoundedness or conditional independence, i.e. that, controlling for observables, the outcomes of the non-treated? control group are independent of innovation (Rosenbaum and Rubin, 1983; Rubin, 2008). Under unconfoundedness, the basic idea is to find in a large group of non-innovative firms similar to the innovative firms in all relevant pre-treatment characteristics X.<sup>10</sup> The match should be based on variables that (i) simultaneously influence the treatment status and the outcome variables (see e.g., Sianesi, 2004; Smith and Todd, 2005); (ii) are unaffected by treatment. Furthermore, over-parameterized models should be avoided (see Bryson et al., 2002) for two reasons, i.e.: (i) including extraneous variables in the propensity score model exacerbates the support problem; (ii) although the inclusion of non-significant variables in the propensity score specification will not bias the propensity score estimates or make them inconsistent, it can increase their variance.

Thus, our final PSM is performed on the following characteristics: *Firm Size, Firm Location, Governance*, and *Portfolio*. While the first three measures represent exogenous firm characteristics, the latter captures the bank evaluation of the firm, as represented by the operating segment into which the firm falls (*Small Business market* or *Corporate market*). Thus, the logistic equation is as follows:

 $\Pr(Innov=1)_{ijt} = F(c + \alpha \ Firm \ Size_{it} + \beta \ Firm \ Location_{it} + \gamma Governance_{t} + \delta \ Portfolio_{it} + \varepsilon_{ijt})_{(3)}$ 

<sup>&</sup>lt;sup>10</sup> Grilli and Rampichini (2011) provide a detailed overview of the most common PSM techniques.

Estimates of the propensity score are shown in Table 3. Then, we match observations on the odds ratio of the propensity score, and once each innovative firm is matched with a non-innovative one, the difference between the outcome of the treated units and the outcome of the matched control units is computed, and the Average Treatment effect on Treated (ATT) of interest is then obtained by averaging these differences. The resulting PSM sample contains 199 innovative and 199 non-innovative firms. As shown in Table 4, the outcome variables (*I\_Rate, Collateral* and *Overdraw*) do not show statistically significant differences after the PSM; therefore, unconfoundedness or a conditional independence assumption turns out to be satisfied, i.e. controlling for observables, the outcomes of the non-treated control group are independent of innovation itself.

#### 5. Results

#### 5.1 Interest rates and Collateral: Simultaneous-equation model

We begin the discussion of our results with the analysis of the results of the simultaneous-equation model (1A)-(1B) shown in Table 5. We discuss findings in columns (1) and (2) that show the estimated coefficients for the PSM sample, while columns (3) and (4) show estimates for the whole sample for comparative purposes.

We note that the coefficient on the variable that captures the innovative firm, namely *Innov*, is positive and statistically significant in columns (1) and negative in column (2). Thus, our preliminary analysis of the impact of innovation on the cost of credit shows that borrowing firms that innovate pay higher interest rates relative to their counterparts supply-dominated firms. This finding is consistent with the part of the theory according to which innovative firms have more difficulty accessing external sources of funds due to riskier activity and more severe asymmetric information problems. Meanwhile, innovative firms usually rely more on intangible assets and therefore, have less tangible assets to use as collateral (Hall, 2010). Further, estimates are in line with previous empirical studies that found that guarantees have a positive impact on interest rates, supporting the idea that observably riskier borrowers are asked to pledge more guarantees to mitigate the moral hazard problem, i.e. interest rates on secured loans are on average higher than those on unsecured loans, and guarantees are most often associated with riskier borrowers and riskier loans (Berger and Udell, 1990 and 1995; Casolaro et al., 2008; Ono and Uesugi, 2009; Ogawa et al., 2010). Indeed, as long as interest rates increase there is an incentive for riskier borrowers to apply for credit (Stiglitz and Weiss, 1981). Therefore, the impact of a high level of collateral on interest rates may be positive if collateral does not counterbalance the higher borrower risk (Calcagnini et al., 2014).

Local market conditions such as competition (HHI) and clusters of economic activity (Industrial District) do not have a statistically significant effect either on the interest rate charged by the bank, nor

on collateral (Table 5, column (1) and (2), respectively). This finding could be motivated by the slowchanging nature of these economic measures and the simultaneous inclusion of bank branch fixed effects (Bellucci *et al.*, 2013).

The legal organization of firms influences both interest rates and collateral requirements. Indeed, we find that firms organized as *Corporations* trade off a higher interest rate against lower collateral requirements. The same finding holds for firms belonging to the bank *corporate market*: the estimated coefficients on *Portfolio* show that credit lines, as part of the bank *corporate market*, have higher interest rates and a lower probability of collateral requirements compared to those that are part of the bank's *small business market*.

Further, larger loans pay a lower price for credit, as they are typically associated with larger firms and they also generate economies of scale (Table 5 column (1)). However, as expected, they have a higher probability of being secured, as shown by the estimated coefficient of the variable *Tot. Accorded* in Table 5, Column (2).

Concerning the variables that aim to capture the bank-borrower relationship, we find that borrowers with longer and exclusive lending relationships do not seem to benefit in terms of price, or collateral requirements, or that the use of other bank services is associated with lower interest rates or collateral. Finally, the *Decisional Level* negatively affects the *price* of credit and positively affects the *nonprice* term of the loan contract, suggesting that the Headquarters decisional level prefers to trade off a lower interest rate with higher collateral requirements.

In short, the positive association between innovation and interest rates is consistent both with an argument for information-based frictions and with the riskier nature of the innovative firm reflected in the *Innov* variable. On the other side, the fact that innovative firms face fewer collateral requirements seems to suggest that in the case of innovative firms banks trade off higher interests rates with lower collateral, likely because of the nature of this type of firms, typically characterized by higher volatility and lower tangible assets than their non-innovative peers.

#### 5.2 Innovation and credit availability

In Table 6 we examine the relation between innovation and credit availability formalized in Equation (2) above. In the first specification (column 1) we estimate a baseline Probit model that includes only our innovative variable, along with industry, year and branch-specific effects. Industry effects are particularly important as some the most critical variables for evaluating the availability of credit are measures of firm risk as leverage or credit score, and Cole (FM, 2013) finds that industry leverage is highly correlated with firm leverage.

In column (2) we augment the baseline specification including a full set of control variables for firm, loan and bank characteristics. In column (3) we exploit the longitudinal dimension of our dataset using panel data estimators to control for unobserved individual effects, which could systematically

differ among entrepreneurs. To this end, we re-estimate our augmented specification from column (2), including random effects. Indeed, our variable of interest, *Innov*, is nearly time-invariant and we cannot use the Within estimator. Lastly, in column (4) we estimate model (1) by using the Hausman and Taylor (1981) estimator (hereinafter, HT). In this procedure, some explanatory variables are allowed to be correlated with the individual effects, and instrumented with the exogenous regressors, the average across time of the exogenous time-variant regressors and the difference between the endogenous time-variant variable and its individual average. Thus, in column (4) *Tot accorded*, *Relationship Length* and *Multiple Lending* are assumed to be endogenous covariates.<sup>11</sup>

The analysis in Table 6 reveals that borrowers that carry out innovative activities are less likely to overdraw funds, as the coefficient of *Innov* is negative and statistically significant in columns (1) and (2). Specifically, the marginal non-tabulated effect of this variable is equal to -0.08, i.e.: being innovative firms reduces the probability of having to face binding credit limits by 8%. Furthermore, we note that our panel estimators – random effects model in column (3) and Hausman-Taylor model in column (4) where *Tot accorded*, *Relationship Length* and *Multiple Lending* are assumed to be endogenous covariates – generate results that confirm the negative association between overdraw and innovation.

Thus, our findings suggest a strong association between firms' innovation activities and credit tightness, and this association seems to be in favor of innovation. Indeed, innovative firms are less likely to overdraw funds, and thus face less binding credit limits than their non-innovative peers.

This finding suggests a dual explanation. First, banks gain the advantage of evaluating the riskier innovative firms, charging them a higher price and lower collateral requirements. Secondly, banks also recognize that innovative firms are more profitable than traditional firms, and "repay" them with less binding credit limits.

While the positive association between interest rates and innovative firms is not new to the empirical literature, to the best of our knowledge, we provide the first comprehensive evidence on the fact that the ability of firms to innovate reduces the probability of overdrawing funds. Thus, firms that innovate have better access to the credit market in *non-price* terms; furthermore banks seem to recognize the opportunity of evaluating the potential high growth and profitability of innovative firms positively.

Turning the analysis of the control variables on the availability of credit (Table 6), we find that local market concentration, as measured by *HHI* reduces the probability of overdrawing. This result is also in line with Audretsch and Elston (2002), in which emerging competition has improved access to capital for some groups of firms. Furthermore, firms organized in more structured legal forms, i.e. *Corporations*, have a lower probability of experiencing binding credit limits, in line with the theory according to which more solid firms have better access to credit markets. Finally, collateralized credit lines present a higher probability of overdrawing funds than unsecured ones.

<sup>&</sup>lt;sup>11</sup> For comparative purposes, Table 7 shows estimates for the full sample.

Table 7 shows estimated coefficients of equation (2) for the full sample, and findings on the impact of innovation on overdraw are confirmed. Furthermore, this Table also shows that the length of the lending relationship has a positive effect on the quantity dimension of the loan contract. Indeed, a longterm banking relationship may benefit the borrowers by helping to build trust between borrowers and lenders (Boot and Thakor, 1994). However, a longer lending relationship might also generate an information monopoly that enables banks to extract rents from firms. Our estimates suggest that the former effect prevails on the latter, i.e.: borrowers with longer bank relationships benefit in terms of credit availability as they are less likely to overdraw costly funds from the bank. Moreover, we find that the use of other bank services is associated with lower probability of overdrawing. Indeed, the scope of the relationship allows banks to accumulate soft information that enhances banks' ability to assess and monitor their customers, and thus to offer better contract terms than other firms. Firms experience a lower probability of overdrawing when their credit lines are managed at the headquarters level, as the indicator Decisional Level is negative and statistically significant. Collateralized credit lines continue to present a higher probability of overdrawing funds than unsecured ones. As expected, the amount of credit reduces the probability of binding credit limits (as shown from the estimated coefficient of Tot. Accorded in columns (2) to (4) in Table 7), credit lines of the corporate market portfolios seem to experience a higher probability of overdraw. The same finding is confirmed for large-sized firms.

To summarize, the positive association between innovation and interest rates is consistent both with an argument for information-based frictions and with the riskier nature of the innovative firm reflected in the *Innov* variable. On the other hand, the fact that innovative firms face less binding credit limits and thus are less likely to overdraw funds seems to be justified by the high levels of profitability and rate of growth that typically are associated with these firms. As we believe that this finding is new and could have interesting policy implications, in the next paragraphs we provide additional evidence on the links between innovation and credit availability.

#### 5.3 Innovation as a Signal: Sub-sample Analyses

The motivation for this study is the recognition that, in terms of access to credit, innovative firms may or may not face more difficulties than traditional ones according whether their weaknesses prevail over their strengths or *vice versa*. On one hand, the innovative nature of their activity implies a higher level of opaqueness and more severe asymmetric information problems. Then, external funds become particularly costly. On the other, innovative firms, as opposed to traditional ones, tend to experience rapid growth rates and the need to attract increasing amounts of external funds, especially risk capital. Our analysis shows that innovative firms pay higher prices than other firms, likely due to their riskiness, and because banks trade off higher interest rates with lower collateral requirements. However, they also face less binding constraints than traditional ones, i.e.: banks consider innovative firms riskier than traditional ones, but assign positive evaluations to the innovative activity they carry out.

As we argue that this finding depends on bank evaluation, in this section we categorize the firms in our sample into sub-samples from a supply-side perspective intended to identify cross-sectional differences in terms of firm transparency. We re-estimate the model of credit availability specified in Model (2) for each sub-sample and draw inferences from these sub-sample analyses shown in Table 8. In each estimation we include the full set of controls used in the estimation of the augmented Probit model in Table 7, column (2).<sup>12</sup>

The first sub-sample categorization considers a measure of the informational advantage of the lending bank for a firm relative to other firms of this bank, i.e. the operating segment type. The sample is divided into two sub-groups: firms that are in the small business portfolio and firms that are in the corporate portfolio. Usually, decisions about the small business portfolio are taken through traditional banking channels at branch premises. By contrast, for firms that fall in the corporate portfolio, bank loan officers usually visit the firm's premises and negotiate the contract terms directly (Bellucci *et al.*, 2013). This sub-sample analysis shows that the coefficient on *Innov* is negative and significant only for firms in the small business portfolio, column (1) of Table 8. Hence, banks evaluate *small businesses* that innovate positively. Meanwhile, when the lender deals with firms in the corporate portfolio, credit availability is not affected by innovation, the decision relies on other firm characteristics.

The second sub-sample division classifies firms according to whether the bank has assigned a credit rating to the firm or not. In the first case, banks rely on additional information about the credit risk of the firm. The findings reveal that firms that innovate face less binding constraints than other firms in the case of they are not rated (4). It follows that, when ratings are available, banks base their decisions on them and are not influenced by the presence or lack of an innovative activity carried out by the firm.

The third part of the sample considers whether the credit line is approved and managed at the local bank branch. In this case, loan officers operating at the branch level may not have enough instruments to evaluate the projects of innovative firms and their potential growth. Hence, one should observe that the importance of innovation in the availability of credit is less pronounced when decisions are taken at a local bank branch. Consistent with this argument, we find that the coefficient on *Innov* is negative and statistically significant only for the sub-sample of firms whose credit lines are managed at the headquarters level (column (5)). These innovative firms are less likely to overdraw funds and exceed their credit limits than traditional ones.

The last sub-sample categorization uses the variable *Relationship Length* to split the sample in two sub-groups: *Short* relationship length (if less than the sample median) and *Long* relationship length (if it exceeds the sample median). The rationale for this partition is that as long as the firm maintains a long-lasting relationship with its bank, loan officers usually base their decisions on soft information accumulated during the years. Diversely, if the firm is a new client or the lending relationship has been

<sup>&</sup>lt;sup>12</sup> Unfortunately, we cannot perform this analysis on the PSM sample because of the poor number of observations we had in each sub sample.

brief, banks may evaluate the information concerning the innovative nature of the firm in a positive manner. Thus, among firms without a well-established lending relationship, those who are considered innovative may face less binding constraints. As expected, the coefficient on *Innov* is negative and statistically significant only for firms who do not have a long-lasting lending relationship, as shown in column (7).

Overall, our sub-sample analyses reveal that innovation is a key determinant for the probability that a firm will experience binding credit limits for those firms that are more likely to suffer from information asymmetry. Among them, banks evaluate positively, in term of credit availability, the presence of an innovative activity carried out by the firm. Meanwhile, for those firms which are well known by the bank and for which the loan officers may rely on additional information, being innovative is not a key determinant. These findings seem to suggest that innovation may reduce the presence of credit constraints among those firms that, typically, may suffer more from financing constraints.

#### 6. Conclusions

This article carried out an empirical analysis of the access to credit by innovative firms compared to traditional ones in *price* and *non-price* terms. Innovative firms may or may not face more difficulties in credit access than traditional ones, according to whether their weaknesses or strengths prevail. In the first case, innovative firms could incur in higher financing costs, and have a higher probability of being credit rationed than traditional firms. In the second case, the situation could be reversed, and innovative firms could get better credit conditions than traditional ones.

Our findings reveal a positive association between innovation and interest rates, and are consistent both with an argument for information-based frictions and with the riskier nature of innovative firms compared to traditional ones. However, we also find that innovative firms face less binding credit limits and thus are less likely to overdraw funds than the others. These findings together suggest that, on one hand, banks evaluate innovative firms to be riskier, and then charge them a higher price.

At the same time, on the other hand, banks also recognize that innovative firms are more profitable than other firms, and reward them with less binding credit limits (Freel, 2007). Our analysis also shows that innovation is a key determinant for the probability that a firm will experience binding credit limits for those firms that are more likely to suffer from problems of information asymmetry. Among them, banks evaluate positively, in term of credit availability, the presence of an innovative activity carried out by the firm. Meanwhile, for those firms which are well known by the bank and for which the loan officers may rely on additional information, being innovative is not a key determinant. These findings seem to suggest that innovation may reduce the presence of credit constraints among those firms that, typically, may suffer more from financing constraints. For these firms, innovation may help to avoid the costly option of overdrawing expensive funds.

In terms of policy implications our findings offer many points of interest. Indeed, one of the motivations for this study was the current economic crisis and the critique made of the Italian banking system as being unable to respond to firms' demand for credit. This is especially true for small firms, that are the ones which suffer most from the crisis (Bank of Italy, 2013). This study reveals that firms that innovate could face lower binding constraints. Therefore, a way to overcome the difficulties of credit access should be to undertake proactive behaviors intended to convert the firm's activity from the perspective of innovation.

Further, the importance of innovative firms for economic growth suggests that all economic and institutional actors should cooperate to increase the number of innovative firms. According to this paradigm banks should financially sustain dynamic and efficient firms, and the public sector should promote industrial policies oriented to growth (Rey, 2012).

#### References

Acs, Z. J., and Audretsch, D. B., (1987). Innovation, market structure, and firm size. The Review of Economics and Statistics 69, 567–574.

Acs, Z. J., and Audretsch, D. B., (1988). Innovation in large and small firms: An empirical analysis. American Economic Review 78, 678-690.

Almus, M. and Czarnitzki, D., (2003). The effects of public R&D subsidies on firms' innovation activities: the case of Eastern Germany. Journal of Business & Economic Statistics 21, 226–36.

Audretsch, D.B, (2002). The Dynamic Role of Small Firms – evidence from the US. Small Business Economics 18, 13-40.

Audretsch, D.B., Boente, W., and Mahagaonkar, P., (2009). Financial Signaling by Innovative Nascent Entrepreneurs. Shumpeter Discussion Papers 2009-001.

Audretsch, D. B., and Elston, J. A., (2002). Does firm size matter? Evidence on the impact of liquidity constraints on firm investment behavior in Germany. International Journal of Industrial Organization 20, 1–17.

Bank of Italy (2013), Annual Report. Rome.

- Bellucci, A., Borisov, A., and Zazzaro, A. (2013). Do banks price discriminate spatially? Evidence from Small Business Lending in local credit markets. Journal of Banking and Finance 37, 4183–4197.
- Benfratello L., Schiantarelli F. and Sembenelli A. (2008), Banks and Innovation: Microeconometric Evidence on Italian Firms, Journal of Financial Economics 90, 197-217.
- Berger, A. N. and Udell, G. F. (1990). Collateral, Loan Quality, and Bank Risk. Journal of Monetary Economics 25, 21-42.
- Berger, A.N. and Udell, G. F. (1995). Relationship Lending and Lines of Credit in Small Firms Finance, Journal of Business 68, 351-381.
- Berger, A.N. and Udell, G. F. (1998) The Economics of Small Business Finance: The Roles of Private Equity and Debt Markets in the Financial Growth Cycle, Journal of Banking & Finance 22, 613-673.
- Berger, A. N., Klapper, L.F. and Udell, G. F. (2001) The ability of banks to lend to informationally opaque small business. Journal of Banking and Finance 25, 2127-2162.
- Berger, A. N. and Udell, G. F. (2002). Small business credit availability and relationship lending: The importance of bank organizational structure. The Economic Journal 112, 32-53.
- Boot, A. and Thakor, A.V. (1994). Moral hazard and secured lending in an infinitely repeated credit market game. International Economic Review 35, 899-920.
- Brick, I.E. and Palia, D. (2007). Evidence of jointness in the terms of relationship lending, Journal of Financial Intermediation 16, 452-476.
- Brierley, P. (2001). The Financing of Technology-based Small Firms. A Review of the Literature, Bank of England Quarterly Bulletin 41, 64-76.

- Brown, M., Degryse, H., Hower, D., and Penas, M.F. (2012). How do banks screen innovative firms? Evidence from start-up panel data. ZEW discussion paper No. 12-032.
- Bryson, A., Dorsett, R., and S. Purdon (2002). The Use of Propensity Score Matching in the Evaluation of Labour Market Policies, Working Paper No. 4, Department for Work and Pensions.
- Calcagnini, G., Favaretto, I. and Giombini, G. (2011). Financial Models of Small Innovative Firms: an Empirical Investigation in The Economics of Small Businesses. An International Perspective, G. Calcagnini e I. Favaretto Edt, serie Contribution to Economics, Springer-Verlag (Berlin Heidelberg).
- Calcagnini, G., Farabullini, I. and Giombini, G. (2014). The impact of guarantees on bank loan interest rates, Applied Financial Economics. Doi: 10.1080/09603107.2014.881967.
- Canepa, A., and Stoneman, P. (2008). Financial constraints to innovation in the UK: evidence from CIS" and CIS3. Oxford economic papers 60, (4) 711-730.
- Carpenter and Petersen, M.A. (2002). Capital market imperfections, high-tech investment and new equity financing. The Economic Journal 112, 54-72.
- Casolaro, L., Focarelli, D. and Pozzolo, A. F. (2008). The pricing effect of certification on syndicated loans, Journal of Monetary Economics 5, 335-349.
- Chava, S., Nanda, V., and Xiao, S.C. (2013). Lending to innovative firms: The role of lender expertise and control rights. WP SSRN 2270740.
- Cole, R., (2013). What Do We Know about the Capital Structure of Privately Held US Firms? Evidence from the Surveys of Small Business Finance. Financial Management 42, 777-813.
- Cole, R., (2010). Bank credit, trade credit or no credit: Evidence from the Surveys of Small Business Finances, MPRA Paper 24689, University Library of Munich, Germany.
- Cole, R. (1998). The importance of relationships to the availability of credit. Journal of Banking and Finance 22, 959-977.
- Cole, R., and Wolken, J. D. (1995). Financial services used by small businesses: evidence from the 1993 National Survey of Small Business Finances, Federal Reserve Bulletin, Board of Governors of the Federal Reserve System (U.S.), issue Jul, 629-667.
- Colombo, M., and Grilli, L. (2007). Funding Gaps? Access To Bank Loans By High-Tech Start-Ups. Small Business Economics 29, 25-46.
- Czarnitzki, D. (2006). Research And Development In Small And Medium-Sized Enterprises: The Role Of Financial Constraints And Public Funding. Scottish Journal of Political Economy 53, 335-357.
- Czarnitzki, D. and Hottenrott, H. (2011). R&D investment and financing constraints of small and medium-sized firms. Small Business Economics 36, 65–83.
- ECB (2009). Survey on the Access to Finance of Small and Medium-Sized Enterprises in the Euro Area. September 2009.
- Favaretto, I. and Zanfei, A. (2007). Rapporto Finale 2007 Innovazione e Centri di Ricerca nelle Marche, Progetto Inter-universitario, Università di Urbino.

Freel, M.S. (2007). Are Small Innovators Credit Rationed?, Small Business Economics, 28: 23-35.

- Fritsch, M., Brixy, U. and Falck, O. (2006). The Effect of Industry, Region and Time on New Business Survival - A Multi-Dimensional Analysis, Review of Industrial Organization 28, pp. 285–306.
- Gorg, H., and Strobl, E. (2007). The Effect of R&D Subsidies on Private R&D. Economica 74, 215–234.
- Grilli, L., and Rampichini, C. (2011). Propensity scores for the estimation of average treatment effects in observational studies. Dipartamento di Statistica "Giuseppe Parenti" Univeriti di Firenze. Training lessons on causal inference. Birstol – June 28-29.
- Available (February 2014) at:
- http://www.bristol.ac.uk/cmm/software/support/workshops/materials/causal-inference/prop-scores.pdf
- Guidici, G. and Paleari, S. (2000). The Provision of Finance to Innovation: A Survey Conducted among Italian Technology-based Small Firms, Small Business Economics 14, 37-53.
- Hajivassiliou, V. and Savignac, F. (2008). Financial constraints and a firm's decision and ability of innovate: Establishing direct and reverse effects. Documents de travail 202. Banque de France.
- Hausman J.A. and Taylor W.E. (1981). Panel data and unobservable individual effects. Econometrica 49, 1377-1398.
- Hall, B., (2010). The financing of Innovative firms. Review of Economics and Institutions 1, 1(4).
- Hall, B., (2002). The Financing of Research and Development, Oxford Review of Economic Policy 18, 35-51.
- Hellmann, T., and Stiglitz, J. (2000). Credit and equity rationing in markets with adverse selection, European Economic Review 44, 281-304.
- Himmelberg, C. P., and Petersen, B.C. (1994). R&D and Internal Finance: A Panel Study of Small Firms in High-Tech Industries, The Review of Economics and Statistics 76, 38-51.

Holmstrom, B., (1989). Agency costs and innovation. Journal of Economic Behavior and Organization, 305–327.

Leuven, E., and Sianesi, B. (2003). psmatch2: Stata module to perform full Mahalanobis and propensity score matching, common support graphing, and covariate imbalance testing. http://ideas.repec.org/c/boc/bocode/s432001.html.

Levin, R. C., Cohen, W. M., and Mowery, D. C. (1985). R & D appropriability, opportunity, and market structure: new evidence on some Schumpeterian hypotheses. The American Economic Review *75*, 20–24.

- Magri, S., (2013). Does Issuing Equity Help R&D Activity? Evidence from Non-Public Italian High-Tech Manufacturing Firms. papers.ssrn.com/abstract=2309909.
- Mocnik, D., (2001). Asset specificit and a firm's borrowing ability: an empirical analysis of manufacturing firms. Journal of Economics Behavior and Organization 45, 69-81.

- OECD (2006). Risk Capital and Innovative SMEs, in The SME Financing Gap (VOL. I): Theory and Evidence, Paris.
- OECD (2011). OECD Science, Technology and Industry Scoreboard. Paris.
- OECD (2011). Financing High-Growth Firms. Paris.
- Ono, A. and Uesugi, I. (2009). The Role of Collateral and Personal Guarantees in Relationship Lending: Evidence from Japan's SME Loan Market. Journal of Money, Credit, and Banking 41, 935-960.
- Petersen, M.A. and Rajan, R.G. (1995). The effect of credit market competition on lending relationships. The Quarterly Journal of Economics 110, 407-443.
- Rey, G. (2012). Un'efficace politica economica è semplice ma non facile. Otto mosse per il rilancio. Finanza, Impresa, Risparmio, Scenari, Tecnologie. FIRST online. 27 marzo.
- Robb, A. (2009). Sources of Financing for New Technology Firms: Evidence from the Kauffman Firm Survey, Kauffman Foundation: Kansas City.
- Rosenbaum, P.R., and Rubin, D.B., (1983). The central role of the propensity score in observational studies for causal effects, Biometrika 70, 41-55
- Rubin, D.B. (2008). For objective causal inference, design trumps analysis, Ann. Appl. Stat., 2, 777-1142.
- Santarelli, E. (2000). The duration of new firms in banking: an application of Cox regression analysis.
- Empirical Economics 25, 315-325.
- Santarelli, E., and Sterlacchini, A. (1990). Innovation, formal vs. informal R&D, and firm size: some evidence from Italian manufacturing firms. Small Business Economics 2, 223–228.
- Savignac, F. (2008). The impact of financial constraints on Innovation: What can be learned from a direct measure? Economics of Innovation and New Technology 17, 553-569.
- Sianesi, B. (2004). An Evaluation of the Active Labour Market Programmes in Sweden, The Review of Economics and Statistics 86, 133–155.
- Smith, J., and Todd, P. (2005). Does Matching Overcome LaLonde's Critique of Nonexperimental Estimators?, Journal of Econometrics 125, 305–353.
- Stiglitz, J., and Weiss, A. (1981). Credit Rationing in Markets with Imperfect Information, American Economic Review 71, 393–410.
- Solow, R.M. (1957). Technical change and aggregate production function. The Review of Economics and Statistics 39, 312-320.
- Zellner, A., and Theil, H. (1962). Three-Stage Least Squares: Simultaneous Estimation of Simultaneous Equations, Econometrica 30, 54-78.

#### TABLE 1-ACTIVITIES OF R&D

ACTIVITY	DESCRIPTION
CORDIS PROJECTS	Community Research and Development Information Service (Cordis) projects that firms located in these provinces participated in this lapse of time.
USPTO PATENTS	Patents registered to United States Patent and Trademark Office (USPTO) by firms located in our referred provinces.
EPO PATENTS	Patents registered to Europena Patent Office (EPO) by firms located in our referred provinces.
UIB PATENTS	Patents registered to Italian Patent Office (UIB) by firms located in our referred provinces.
NATIONAL REGISTRY OFFICE	Firms registered at "Anagrafe Nazionale delle Ricerche" (ANR). ANR is a national registry office where firms registration' is mandatory obtaining public funds for R&D.
FAR PROJECTS	Projects funded by the Ministry of Education, University and Research (MIUR) through the "Fondo Agevolazioni alla Ricerca" (FAR). FAR is a public national fund supporting firms national projects' in R&D following the laws 299/97, 488, ex 1.46.
REGISTER OF LABORATORIES	Register of Laboratories managed by the Ministry of Education, University and Research (MIUR). Firms are registered only after a selective evaluation of activities by a specific departmental office.
REGIONAL PROJECTS	Firms projects' financed by regional funds to incentive R&D with regional law 73°, 73b/2003.

	All	All firms Innovative Firms		Traditio	nal Firms	Means differences	
	Mean	St. dev.	Mean	St. dev.	Mean	St. dev.	p-values
Dependent variables	5						1
Interest rate	7.043	(2.470)	6.588	(2.445)	7.050	(2.470)	0.004***
Overdraw	0.220	(0.414)	0.061	(0.240)	0.222	(0.416)	0.000***
Collateral	0.301	(0.459)	0.175	(0.381)	0.303	(0.459)	0.000***
Innovation variables	s						
Innov	0.015	(0.123)					
Control variables							
Industrial							
district	0.585	(0.493)	0.795	(0.405)	0.582	(0.493)	0.000 ***
HHI	0.213	(0.155)	0.165	(0.116)	0.213	(0.155)	0.000 ***
Sales	2.256	(1.673)	4.876	(1.735)	2.215	(1.639)	0.000 ***
Corporation	0.339	(0.473)	0.923	(0.267)	0.330	(0.470)	0.000 ***
Multiple		. ,				. ,	
Lending	0.960	(0.195)	0.961	(0.193)	0.960	(0.195)	0.917
Other Services	0.911	(0.284)	0.974	(0.158)	0.910	(0.286)	0.001 ***
Relationship							
Length	3380	(2724)	5089	(3219)	3353	(3708)	0.000 ***
Portfolio	0.103	(0.304)	0.594	(0.492)	0.095	(0.293)	0.000 ***
Decisional							
Level	0.178	(0.383)	0.581	(0.494)	0.172	(0.377)	0.000 ***
Credit Limit	117,234	(595,432)	1,008,559	(2,805,378)	103,538	(476,641)	0.000 ***
Credit Used	80,120	(392,907)	542,707	(1,710,773)	72,992	(329,393)	0.000 ***
Rating	5.675	(1.512)	4.923	(2.104)	5.704	(1.478)	0.000 ***

#### TABLE 2 - SUMMARY STATISTICS: INNOVATIVE VS. TRADITIONAL FIRMS

#### TABLE 3 - PROPENSITY SCORE MATCHING: LOGISTIC ESTIMATION

Dependent variable: Innov	Coef.	Std. Err.	Z	$P>_Z$	[95% Conf.	Interval]
Portfolio	0.55	0.22	2.53	0.01	0.12	0.98
D (Governance 2)	2.39	0.46	5.22	0.00	1.50	3.29
D (Governance 3)	1.09	0.51	2.12	0.03	0.08	2.09
D (Governance 4)	1.49	0.55	2.69	0.01	0.41	2.58
D (Governance 5)	3.08	0.50	6.12	0.00	2.09	4.07
Province	-2.37	0.42	-5.67	0.00	-3.18	-1.55
D (Sales 2)	0.42	0.51	0.82	0.42	-0.58	1.42
D (Sales 3)	0.96	0.39	2.48	0.01	0.20	1.71
D (Sales 4)	1.79	0.36	5.04	0.00	1.10	2.49
D (Sales 5)	1.94	0.39	5.02	0.00	1.18	2.70
D (Sales 6)	2.13	0.43	4.93	0.00	1.28	2.97
Constant	-7.07	0.44	-16.13	0.00	-7.92	-6.21
Observations		14376				
LR Chi $2(12)$ p value		0.00				
Pseudo R2		0.2632				

*Note:* The table presents results from the PSM logistic regression (3). PSM is performed by using the psmatch2 (Leuven and Sianesi 2003) Stata command. We used the option nearest neighbor (within caliper, without replacement), and imposing common support. The definition and construction of the variables is provided in the Appendix. \* p<0.1, \*\* p<0.05, \*\*\* p<0.01.

Outcome Variable	Sample	Treated	Controls	Difference	S.E.	T-stat
	-	(Innov=1)	(Innov=0)			
I_rate	Unmatched	6.87	7.07	-0.20	0.17	-1.15
	ATT	6.87	6.64	0.24	0.23	1.03
Collateral	Unmatched	0.20	0.31	-0.12	0.03	-3.51
	ATT	0.20	0.22	-0.02	0.04	-0.49
Overdraw	Unmatched	0.06	0.23	-0.17	0.03	-5.56
	ATT	0.06		-0.05	0.03	-1.80

*Note*: The table presents outcomes (*I\_Rate, Collateral,* and *Overdram*) before and after the PSM computed according to equation (3). The "unmatched sample" refers to the mean value of outcome variables before the match. The "ATT" represents the mean value of outcome variables after the match.

	Matchee	d sample	Full sample		
VARIABLES	I_Rate	Collateral	I_Rate	Collatera	
To a ser	0.(2*	0.00*	0.27*	0.10*	
Innov	0.62*	-0.09*	0.37*	-0.10*	
	(0.371)	(0.050) $0.14^{***}$	(0.205)	(0.056)	
[_Rate				0.27***	
	( 70***	(0.041)	2 ( 5 4 4 4	(0.003)	
Collateral	6.78***		3.65***		
	(2.044)	0.02	(0.034)	0.05*	
Industrial District	0.17	-0.02	0.17*	-0.05*	
11 11	(0.539)	(0.080)	(0.091)	(0.025)	
HHI	1.83	-0.27	0.37	-0.10	
	(2.322)	(0.339)	(0.368)	(0.101)	
Corporation	1.28*	-0.18*	0.93***	-0.25***	
	(0.688)	(0.105)	(0.059)	(0.016)	
Distance	0.13	-0.02	0.06***	-0.02***	
	(0.134)	(0.019)	(0.017)	(0.005)	
Multiple Lending	0.77	-0.11	0.02	-0.01	
	(1.029)	(0.146)	(0.132)	(0.036)	
Other Services	0.33	-0.05	0.21**	-0.06***	
	(0.996)	(0.145)	(0.083)	(0.023)	
Relationship Length	0.02	-0.00	-0.02	0.00	
r C	(0.173)	(0.025)	(0.025)	(0.007)	
Portfolio	1.44**	-0.21***	0.13	-0.04	
	(0.605)	(0.073)	(0.123)	(0.034)	
Decisional Level	-1.23**	0.18**	0.01	-0.00	
	(0.568)	(0.078)	(0.085)	(0.023)	
l'otal accorded	-0.70***	0.10***	-0.66***	0.18***	
	(0.217)	(0.026)	(0.025)	(0.007)	
Province	0.49	-0.08	0.38	-0.10	
lovince	(2.979)	(0.429)	(0.284)	(0.078)	
D (Sales 2)	-0.94	0.14	0.08	-0.02	
(Sales 2)	(1.102)	(0.160)	(0.081)	(0.022)	
D (Sales 3)	1.09	-0.16	0.45***	-0.12***	
(Sales 5)	(0.932)	(0.130)			
$\mathcal{O}(Salas A)$	2.36**	-0.34***	(0.078) 1.05***	(0.021) -0.29***	
D (Sales 4)					
D (8-1 F)	(0.944)	(0.125)	(0.093) 1.00***	(0.025) -0.28***	
D (Sales 5)	0.54	-0.08			
	(0.883)	(0.124)	(0.131)	(0.036)	
D (Sales 6)	0.18	-0.03	0.89***	-0.24***	
	(0.944)	(0.136)	(0.206)	(0.056)	
Constant	10.84***	-1.56***	11.87***	-3.26***	
	(3.472)	(0.600)	(0.426)	(0.119)	
Branch effects	Yes	Yes	Yes	Yes	
l'ime effects	Yes	Yes	Yes	Yes	
Industry effects	Yes	Yes	Yes	Yes	
Observations	390	390	14,501	14,501	
R-squared	-0.22	-0.82	-1.61	-0.25	
Hansen-Sargan overid. Statistic (p-value)	0.72		0.82		

## TABLE 5 – INNOVATION, INTEREST RATE AND COLLATERAL: SIMULTANEOUS-EQUATION ESTIMATES

*Note:* The table presents simultaneous-equation analysis of the impact of innovation on *Interest Rate* and *Collateral* (equations (1A-1B)) by using the reg3 Stata command. Columns (1) and (2) show results from from PSM sample, while columns (3) and (4) show estimates the whole sample. Endogenous variables are instrumented by *Rival Distance*, *Rating*, and *Overdraw* in the first step estimates. Instrument validity is tested by using the overid Stata command after estimation. The table reports point estimates of the coefficients, followed in parentheses by t-statistics, based on robust standard errors. The definition and construction of the variables is provided in the Appendix. \* p<0.1, \*\* p<0.05, \*\*\* p<0.01

	Overdraw	Overdraw	Overdraw	Overdraw
VARIABLES	(1)	(2)	(3)	(4)
Innov	-0.40**	-0.59***	-0.07**	-0.08*
	(0.172)	(0.192)	(0.031)	(0.044)
Industrial District		0.47	0.04	0.03
		(0.502)	(0.050)	(0.068)
HHI		-9.75***	-0.36*	-0.37*
		(2.744)	(0.185)	(0.196)
Corporation		-1.22**	-0.17	-0.16*
1		(0.574)	(0.105)	(0.086)
Multiple Lending		-0.76*	-0.07	0.00
1 0		(0.432)	(0.078)	(0.116)
Other Services		0.53	0.11	0.00
		(0.775)	(0.099)	(0.139)
Relationship Length		-0.11	-0.01	0.05
		(0.160)	(0.025)	(0.045)
Portfolio		0.06	0.00	-0.03
		(0.271)	(0.059)	(0.069)
Decisional Level		0.20	-0.01	-0.07
		(0.495)	(0.090)	(0.078)
Tot. Accorded		-0.19	-0.03	0.00
		(0.188)	(0.027)	(0.041)
Distance		0.08	0.00	-0.01
		(0.155)	(0.015)	(0.014)
D (Collateral 2)		0.20	0.07*	0.04
		(0.287)	(0.039)	(0.048)
D (Collateral 3)		1.77**	0.34*	0.29***
		(0.751)	(0.183)	(0.091)
D (Sales 2)		0.46	-0.01	-0.06
		(0.880)	(0.179)	(0.139)
D (Sales 3)		0.06	0.03	-0.03
		(0.603)	(0.109)	(0.115)
D (Sales 4)		0.58	0.08	0.01
		(0.568)	(0.109)	(0.108)
D (Sales 5)		0.38	0.07	0.01
		(0.537)	(0.104)	(0.109)
D (Sales 6)		0.06	0.02	-0.05
		(0.712)	(0.111)	(0.118)
Province		1.52*	-0.05	-0.12
		(0.904)	(0.133)	(0.328)
Constant	-1.39***	3.64	0.51	0.09
	(0.189)	(2.883)	(0.371)	(0.641)
Branch effects	Yes	Yes	Yes	Yes
Year effects	Yes	Yes	Yes	Yes
Industry effects	Yes	Yes	Yes	Yes
Observations	283	283	396	396

#### TABEL 6 - INNOVATION AND OVERDRAW: PSM SAMPLE

Note: The table presents multivariate analysis of the impact of innovation on *Overdraw* (Model (2)). Column (1) shows results from a baseline Probit model with dependent variable *Overdraw* which takes the value of 1 if the firm exceeds the amount granted on the credit line by the bank and 0 otherwise. Column (2) shows results from an augmented Probit model. Column (3) shows results from a Random Effects model, while column (4) shows estimates from the Hausman and Taylor (1981) estimator with *Relationship Length, Multiple Lending*, and *Tot. Accorded* as endogenous variables. The table reports point estimates of the coefficients, followed in parentheses by t-statistics, based on robust standard errors. The definition and construction of the variables is provided in the Appendix. \* p < 0.05, \*\*\* p < 0.01

TABLE 7 INNOVATION AND CRED					
	Overdraw	Overdraw	Overdraw	Overdraw	
VARIABLES	(1)	(2)	(3)	(4)	
nnov	-0.58***	-0.30**	-0.03	-0.03	
mov	(0.126)	(0.133)	(0.019)	(0.036)	
ndustrial District	(0.120)	0.05	( /	· · ·	
ndustrial District			0.02	0.02	
11 11		(0.056)	(0.015)	(0.016)	
IHI		-0.06	-0.03	-0.03	
		(0.109)	(0.029)	(0.042)	
Corporation		-0.05	-0.02	-0.01	
		(0.040)	(0.010)	(0.010)	
Distance		0.02	0.01*	0.01**	
		(0.011)	(0.003)	(0.003)	
Iultiple Lending		-0.05	-0.00	0.01	
		(0.072)	(0.017)	(0.022)	
Other Services		-0.12***	-0.05***	-0.06***	
		(0.045)	(0.016)	(0.014)	
elationship Length		-0.11***	-0.03***	-0.02***	
		(0.014)	(0.004)	(0.007)	
ortfolio		0.21**	0.06***	0.06***	
		(0.089)	(0.020)	(0.021)	
Decisional Level		-0.11***	-0.02**	-0.02	
		(0.039)	(0.010)	(0.013)	
'ot. Accorded		-0.19***	-0.05***	-0.05***	
		(0.013)	(0.004)	(0.004)	
O (Collateral 2)		0.23***	0.05***	0.05***	
		(0.028)	(0.008)	(0.009)	
(Collateral 3)		0.59***	0.17***	0.16***	
		(0.045)	(0.014)	(0.012)	
O (Sales 2)		0.14***	0.03***	0.03**	
(outeo 2)		(0.041)	(0.011)	(0.014)	
(Sales 3)		0.11**	0.03**	0.03*	
(Sales 5)		(0.051)	(0.012)	(0.013)	
O (Sales 4)		0.23***	0.06***	0.06***	
(Sales 4)		(0.056)	(0.014)	(0.016)	
(Salaa E)		0.15	0.04**	0.04*	
(Sales 5)					
		(0.096)	(0.020)	(0.023)	
O (Sales 6)		-0.12	0.00	0.00	
		(0.130)	(0.025)	(0.036)	
rovince		0.21***	0.07***	0.07	
		(0.050)	(0.017)	(0.049)	
Constant	-0.77***	2.03***	0.97***	0.91***	
	(0.067)	(0.234)	(0.061)	(0.075)	
ranch effects	Yes	Yes	Yes	Yes	
'ear effects	Yes	Yes	Yes	Yes	
ndustry effects	Yes	Yes	Yes	Yes	
Observations	14,756	14,695	14,695	14,695	
Pseudo) R-squared	0.03	0.08	0.08		

ETHT CAMPLE ANIALVEIC

TADIE

ININOVATION AND ODDOTT AVAILABLE TV

Note: The table presents multivariate analysis of the impact of innovation on *Overdraw* (Model (2)). Column (1) shows results from a baseline Probit model with dependent variable *Overdraw* which takes the value of 1 if the firm exceeds the amount granted on the credit line by the bank and 0 otherwise. Column (2) shows results from an augmented Probit model. Column (3) shows results from a Random Effects model, while column (4) shows estimates from the Hausman and Taylor (1981) estimator. The table reports point estimates of the coefficients, followed in parentheses by t-statistics, based on robust standard errors. The definition and construction of the variables is provided in the Appendix. \* p<0.1, \*\* p<0.05, \*\*\* p<0.01

	Por	tfolio	Rat	ing	Decisio	nal Level		onship 1gth
	SBs	Corporate	Rating	No Rating	HQs	Branch	Short	Long
VARIABLES	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Innov	-0.36*	-0.20	0.01	-0.40***	-0.39*	-0.28	-0.46**	-0.02
milov	(0.202)	(0.298)	(0.316)	(0.144)	(0.224)	(0.181)	(0.227)	(0.227)
Industrial District	0.06	0.04	0.22	0.05	-0.09	0.07	0.07	-0.02
industrial District	(0.059)	(0.111)	(0.215)	(0.063)	(0.173)	(0.060)	(0.059)	(0.123)
HHI	-0.04	-0.46	-56.84***	-0.09	-0.10	-0.05	-0.18	0.22
	(0.116)	(0.604)	(6.478)	(0.105)	(0.436)	(0.125)	(0.132)	(0.204)
Corporation	-0.03	-0.39*	-0.12	-0.06	-0.00	-0.05	-0.02	-0.24**
Solpolation	(0.035)	(0.210)	(0.116)	(0.041)	(0.123)	(0.042)	(0.045)	(0.094)
Distance	0.02	-0.01	-0.06**	0.02**	-0.00	0.02	0.02*	-0.00
	(0.012)	(0.061)	(0.025)	(0.012)	(0.031)	(0.012)	(0.011)	(0.024)
Multiple Lending	-0.08	0.48*	0.16	-0.08	-0.12	-0.04	0.04	-0.21*
	(0.077)	(0.282)	(0.174)	(0.079)	(0.111)	(0.096)	(0.097)	(0.114)
Other Services	-0.12**	-0.12	0.25	-0.13***	0.10	-0.13***	-0.15***	0.02
	(0.046)	(0.213)	(0.316)	(0.044)	(0.157)	(0.047)	(0.047)	(0.162)
Relationship Length	-0.12***	-0.02	-0.08	-0.11***	-0.09*	-0.11***	(01011)	(01202)
	(0.014)	(0.055)	(0.101)	(0.014)	(0.047)	(0.012)		
Portfolio	(0.01.)	(01000)	0.09	0.25**	-0.18	0.36***	0.15	0.46***
			(0.172)	(0.109)	(0.131)	(0.080)	(0.111)	(0.163)
Decisional Level	-0.06	-0.20	-0.33*	-0.07*	(01101)	(01000)	-0.08*	-0.24***
	(0.041)	(0.267)	(0.175)	(0.041)			(0.046)	(0.088)
Tot. Accorded	-0.20***	-0.20***	-0.18***	-0.20***	0.02	-0.21***	-0.20***	-0.17***
	(0.012)	(0.075)	(0.054)	(0.013)	(0.057)	(0.013)	(0.013)	(0.034)
D (Collateral 2)	0.24***	0.10	0.14	0.26***	0.20**	0.25***	0.21***	0.23***
	(0.031)	(0.249)	(0.088)	(0.029)	(0.083)	(0.029)	(0.037)	(0.070)
D (Collateral 3)	0.60***	0.28	0.96***	0.59***	0.52***	0.61***	0.56***	0.80***
	(0.046)	(0.249)	(0.202)	(0.047)	(0.126)	(0.047)	(0.053)	(0.087)
Province	0.23***	-0.22**	5.63***	0.16***	-0.11	0.22***	0.24***	0.13
	(0.069)	(0.113)	(0.412)	(0.053)	(0.305)	(0.077)	(0.068)	(0.105)
Constant	2.11***	1.63	10.27***	2.13***	-0.95	2.13***	1.26***	0.94*
	(0.217)	(1.410)	(1.341)	(0.232)	(1.134)	(0.234)	(0.224)	(0.520)
Branch effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Firm Size effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	13,363	1,107	1,314	13,339	2,323	12,304	11,008	3,718
Pseudo R-squared	0.08	0.16	0.13	0.08	0.08	0.07	0.07	0.09

#### TABLE 8 INNOVATION AND OVERDRAW – SUB-SAMPLE ANALYSIS

Note: The table presents multivariate analysis of the impact of innovation on *Overdraw* (Model (3)). Columns (1) and (2) show results for the sub-samples of firms considered as part of the Small Business portfolio and Corporate portfolio, respectively. Columns (3) and (4) show results for the sub-samples of firms that are rated or not by the bank, respectively. Columns (5) and (6) show results for the sub-samples of cases where the credit line is managed at the headquarters and at the branch level, respectively. Columns (7) and (8) show results for the sub-samples of firms with relationships shorter and longer, respectively, than the 75th percentile of the bank-firm relationships in the sample. Each regression uses the entire sample period and includes industry, time and bank branch effects. The table reports point estimates of the coefficients, followed in parentheses by t-statistics, based on robust standard errors. The definition and construction of the variables is provided in the Appendix. \* p<0.1, \*\* p<0.05, \*\*\* p<0.01

### Appendix: List of Variables

Variable	Definition
Innovative	A dichotomous variable that take the value of 1 if the firm is involved in many of innovative activities following the classification specified in Table 1, and 0 otherwise.
Interest Rate	The interest rate charged by the bank, expressed as a percentage.
Overdraw	A dichotomous variable that takes the value of 1 if the borrower uses more than the amount granted on the credit line by the bank and 0 otherwise.
Collateral	A dichotomous variable that takes the value of 1 if the borrower's credit line is collateralized by real guarantees and 0 otherwise.
Sales	A step variable that takes the value of: It takes the value of 1 if sales are less than €250,000; 2 for sales between €250,000 and €500,000; 3 for sales between €500,000 and €1,500,000; 4 for sales between €1,500,000 and €5,000,000, 5 for sales between €5,000,000 and €25,000,000; 6 for sales between €25,000,000 and €50,000,000.
D(Sales i)	An indicator variable that takes the value of 1 if the firm's sales fall in the <i>i</i> -th category (1 through 6) and 0 otherwise.
Multiple lending	A dichotomous variable that takes the value of 1 if the borrower has multiple bank- borrower relationships and 0 if the borrower has an exclusive relationship with this bank.
Other services	A dichotomous variable that takes the value of 1 if the bank branch provides other (besides the credit line) services to the borrower and 0 otherwise.
Relationship Length	A continuous variable that measures the length of the bank-borrower relationship: constructed as the natural logarithm of 1 + the length of the bank-borrower relationship expressed in days.
Decisional Level	A dichotomous variable that takes the value of 1 if the credit line is managed at the headquarters level and 0 if this happens at a local bank branch.
Amount Collateral	A step variable that takes the value of 1 if the credit line is not secured with collateral, 2 if the line is secured with collateral of less than 80% of the credit line ( <i>Medium</i> <i>collateral</i> ), and 3 if the collateralization is higher than 80% ( <i>High collateral</i> ). Indicator for each collateralization category is used in the multivariate analysis.
D(Collateral i)	An indicator variable that takes the value of 1 if the collateralization of the credit line is in the <i>i</i> -th category (1 through 3) and 0 otherwise.
Portfolio	A dichotomous variable that takes the value of 1 if the bank considers the credit line as part of its <i>small business market</i> and 0 if it is part of its <i>corporate market</i> .
Industrial District	A dichotomous variable that takes the value of 1 if the firm is located within an industrial district area and 0 otherwise.
Corporate	A dichotomous variable that takes the value of 1 if the loan recipient is a corporation and 0 otherwise
Rating	A dichotomous variable that takes the value of 1 if the firm is rated by the bank; 0 if the firm has no rating evaluation.
HHI	A branch-based Herfindahl-Hirschman Index of market concentration. The relevant market for each bank branch is determined by the postal area code where the branch is located.
Tot. Accorded	A continuous variable that measures the total amount accorded: constructed as the natural logarithm of 1+ total amount accorded to the firm.
Governance	This variable stores the firm legal structure into classes. It takes the value of 1 if the firm is organized as Sole proprietorship; the value of 2 if the firm is a Limited liability company; the value of 3 if it is a Partnership; the value of 4 if it is a Limited partnership; the value of 5 if it is a Limited company.
D (Governace i)	An indicator variable that takes the value of 1 if the firm's governance falls in the $i$ -th category (1 through 5) and 0 otherwise.

Distance	A continuous variable that measures the bank-borrower distance: constructed as the natural logarithm of $1 +$ the length of the bank-borrower distance in meters.
Rival distance	Natural logarithm of the 25th percentile of the metric distances between borrower and bank branches of competing banks in the local credit market

## IAW-Diskussionspapiere

Die IAW-Diskussionspapiere erscheinen seit September 2001. Die vollständige Liste der IAW-Diskussionspapiere von 2001 bis 2011 (Nr. 1-77) finden Sie auf der IAW-Internetseite www.iaw.edu/publikationene/iaw-diskussionspapiere.

#### IAW-Diskussionspapiere seit 2011:

Nr. 78 The Role of Employees for Post-Entry Firm Growth Andreas Koch / Jochen Späth / Harald Strotmann	(Februar 2012)
Nr. 79 Nationality Matters: The Geographic Origin of Multinationals and the Productivity of their Foreign Affiliates Christian Arndt / Julia Spies	(Februar 2012)
Nr. 80 All You Need Is Trade: On the In(ter)dependence of Trade and Asset Holdings in Gravity Equations Jörn Kleinert / Katja Neugebauer	(März 2012)
Nr. 81 The Tradeoff between Redistribution and Effort: Evidence from the Field and from the Lab Claudia M. Buch / Christoph Engel	(März 2012)
Nr. 82 Factor Shares and Income Inequality – Empirical Evidence from Germany 2002-2008 Martin Adler / Kai Daniel Schmid	(Mai 2012)
Nr. 83 Network and Selection in International Migration to Spain Nina Neubecker, Marcel Smolka, Anne Steinbacher	(Mai 2012)
Nr. 84 Do Better Capitalized Banks Lend Less? Long-Run Panel Evidence from Germany Claudia M. Buch / Esteban Prieto	(Mai 2012)
Nr. 85 From the Stability Pact to ESM – What next? Claudia M. Buch	(Juni 2012)
Nr. 86 The Connection between Imported Intermediate Inputs and Exports: Evidence from Chinese Firms Ling Feng / Zhiyuan Li / Deborah L. Swenson	(Juni 2012)
Nr. 87 EMU and the Renaissance of Sovereign Credit Risk Perception Kai Daniel Schmid / Michael Schmidt	(August 2012)
Nr. 88 The Impact of Random Help on the Dynamics of Indirect Reciprocity Charlotte Klempt	(September 2012)
Nr. 89 Specific Measures for Older Employees and Late Career Employment Bernhard Boockmann / Jan Fries / Christian Göbel	(Oktober 2012)
Nr. 90 The Determinants of Service Imports: The Role of Cost Pressure and Financial Constraints Elena Biewen / Daniela Harsch / Julia Spies	(Oktober 2012)
Nr. 91 Mindestlohnregelungen im Maler- und Lackiererhandwerk: eine Wirkungsanalyse Bernhard Boockmann / Michael Neumann / Pia Rattenhuber	(Oktober 2012)
Nr. 92 Turning the Switch: An Evaluation of the Minimum Wage in the German Electrical Trade Using Repeated Natural Experiments Bernhard Boockmann / Raimund Krumm / Michael Neumann / Pia Rattenhuber	(Dezember 2012)

# IAW-Diskussionspapiere

Nr. 93 Outsourcing Potentials and International Tradability of Jobs Evidence from German Micro-Level Data Tobias Brändle / Andreas Koch	(Januar 2013)
Nr. 94 Firm Age and the Demand for Marginal Employment in Germany Jochen Späth	(Februar 2013)
Nr. 95 Messung von Ausmaß, Intensität und Konzentration des Einkommens- und Vermögensreichtums in Deutschland Martin Rosemann / Anita Tiefensee	(Juli 2013)
Nr. 96 Flexible Collective Bargaining Agreements: Still a Moderating Effect on Works Council Behaviour? Tobias Brändle	(Oktober 2013)
Nr. 97 New Firms and New Forms of Work Andreas Koch / Daniel Pastuh / Jochen Späth	(Oktober 2013)
Nr. 98 Non-standard Employment, Working Time Arrangements, Establishment Entry and Exit Jochen Späth	(November 2013)
Nr. 99 Intraregionale Unterschiede in der Carsharing-Nachfrage – Eine GIS-basierte empirische Analyse Andreas Braun / Volker Hochschild / Andreas Koch	(Dezember 2013)
Nr. 100 Changing Forces of Gravity: How the Crisis Affected International Banking Claudia M. Buch / Katja Neugebauer / Christoph Schröder	(Dezember 2013)
Nr. 101 Vertraulichkeit und Verfügbarkeit von Mikrodaten Gerd Ronning	(Januar 2014)
(Januar 2014) Vermittlerstrategien und Arbeitsmarkterfolg: Evidenz aus kombinierten Prozess- und Befragungsdaten Bernhard Boockmann / Christopher Osiander / Michael Stops	
Nr. 103 Evidenzbasierte Wirtschaftspolitik in Deutschland: Defizite und Potentiale Bernhard Boockmann / Claudia M. Buch / Monika Schnitzer	(April 2014)
Nr. 104 Does Innovation Affect Credit Access? New Empirical Evidence from Italian Small Business Lending Andrea Bellucci / Ilario Favaretto / Germana Giombini	(Mai 2014)